Author's response to reviews

Title: HIV-associated neurocognitive disorders in sub-Saharan Africa: a pilot study in Cameroon.

Authors:

Georgette D Kanmogne (gkanmogne@unmc.edu)
Callixte T Kuate (kuate_callixte@hotmail.com)
Lucette A Cysique (lcysique@unsw.edu.au)
Julius Y Fonsah (funsah@yahoo.co.uk)
Sabine Eta (sabeta2001@yahoo.fr)
Roland Doh (land2000ca@yahoo.co.uk)
Dora M Njamnshi (dmnjamnshi@yahoo.co.uk)
Emilienne Nchindap (eminchindap4@yahoo.com)
Donald R Franklin Jr (dofranklin@ucsd.edu)
Ronald J Ellis (roellis@ucsd.edu)
Allen J McCutchan (amccutchan@ucsd.edu)
Fidele Binam (binam_04nj@ymail.com)
Dora Mbanya (dmbanya1@yahoo.co.uk)
Robert K Heaton (rheaton@ucsd.edu)
Alfred K Njamnshi (aknjamnshi@yahoo.co.uk)

Version: 3 Date: 25 June 2010

Author's response to reviews:

We have made the following- requested changes:

1. We revised the author list to make sure all authors names conform to the journal specifications. Thus, J. Allen McCutchan was changed to John A McCutchan.

2. All sections in the revised manuscript are in black.

3. We removed the vertical line on Table 5. Thus, there is no vertical line in any of the tables in the revised manuscript.